Author's response to reviews

Title: Incidence and determinants of severe maternal morbidity: a transversal study in a referral hospital in Teresina, Piaui, Brazil

Authors:

Alberto P Madeiro (madeiro@uol.com.br)
Andrea C Rufino (andreacrufino@gmail.com)
Erica ZG Lacerda (ezg.lacerda@yahoo.com.br)
Lais G Brasil (lag.brasil@bol.com.br)

Version: 7
Date: 16 August 2015

Author's response to reviews:

1. Please confirm whether the institutional address provided corresponds to all authors, as currently there is no link between the author list and institutional address (there is no association between this address and the number placed next to each author's name)

We did the association between authors and institutional address (Page 1).

2. Please confirm whether informed consent was obtained. If the need for informed consent was waived by an IRB or whether consent is deemed unnecessary according to national regulations, please clearly state this in the manuscript and provide the IRB name or reference to relevant legislation.

The informed consent was waived by Institutional Review Board. We added this information (Methods Section, page 6, lines 150-151).